Author's response to reviews

Title: Adenosine A2b Receptor Promotes Progression of Human Oral Cancer

Authors:

Hiroki Kasama (porsche_0108@yahoo.co.jp)
Yosuke Sakamoto (end@faculty.chiba-u.jp)
Atsushi Kasamatsu (kasamatsua@faculty.chiba-u.jp)
Atsushi Okamoto (onefoot_mactwist@yahoo.co.jp)
Tomoyoshi Koyama (koyamatomoyo4@gmail.com)
Yasuyuki Minakawa (d02a032d@yahoo.co.jp)
Katsunori Ogawara (katsunori.ogawara@faculty.chiba-u.jp)
Hidetaka Yokoe (yokoe@ndmc.ac.jp)
Masashi Shiiba (m.shiiba@faculty.chiba-u.jp)
Hideki Tanzawa (tanzawap@faculty.chiba-u.jp)
Katsuhiro Uzawa (uzawak@faculty.chiba-u.jp)

Version: 2 Date: 20 September 2014

Author's response to reviews:

Dr. Dafne Solera
Executive Editor, BMC cancer
BioMed Central
Floor 6, 236 Gray's Inn Road
London
WC1X 8HB
United Kingdom

Dear Dr. Dafne Solera,

We are now sending our revised manuscript entitled ‘Adenosine A2b Receptor Promotes Progression of Human Oral Cancer’ by Kasama et al for consideration for publication in BMC cancer.

We have revised the manuscript according to the requests from editor (Line Numbering, Title page, Ethics).

We would be grateful if the manuscript is reviewed and considered for publication in BMC cancer.

Thank you very much for your special consideration.

Respectfully yours,

Katsuhiro Uzawa, DDS, PhD
Associate Professor
Department of Oral Science
Graduate School of Medicine
Chiba University
1-8-1 Inohana, Chuo-ku
Chiba 260-8670, Japan
Phone&Fax: +81-43-226-2151
E-mail: uzawak@faculty.chiba-u.jp